Author's response to reviews

Title: Identification of Transcriptional Regulatory Networks Specific to Pilocytic Astrocytoma

Authors:

Hrishikesh Deshmukh (hdeshmuk@gmail.com)
Jinsheng Yu (JYu@path.wustl.edu)
Jahangheer Shaik (ishaik@path.wustl.edu)
Tobey J MacDonald (tmacdon@cnmc.org)
Arie Perry (aperry@pathology.wustl.edu)
Jacqueline E Payton (jpayton@wustl.edu)
David H Gutmann (gutmannd@neuro.wustl.edu)
Mark A Watson (watsonm@wustl.edu)
Rakesh Nagarajan (rakesh@wustl.edu)

Version: 5 Date: 15 March 2011

Author's response to reviews:

March 14, 2011

To the Editors,

Thank you for consideration and acceptance of our manuscript entitled, “Identification of Transcriptional Regulatory Networks Specific to Pilocytic Astrocytoma”. Per your request, we have added (and highlighted) language in our Materials and Methods section that describes compliance with use of human biospecimens, according to our Institutional Review (Ethics) Board. We have also made minor changes and additions (highlighted) to conform with your editorial policies and formatting template. Note that for purposes of brevity, we have combined our Results and Discussion sections. If this is not an acceptable format, please let us know. If you require any additional information from us, please do not hesitate to contact us.

Sincerely –

Mark A. Watson